Author's response to reviews

Title: Efficacy and Safety of Fermented Turmeric Powder in Subjects with Elevated Alanine Transaminase Levels

Authors:

Sang-Uk Kim (clickm@chonbuk.ac.kr)
Ki-Chan Ha (omphalos@chonbuk.ac.kr)
Eun-Kyung Choi (ekchoi@jbctc.org)
Min-Gul Kim (mgkim@jbctc.org)
Dae-Young Kwon (dykwon@kfri.re.kr)
Hye-Jeong Yang (yhi@kfri.re.kr)
Min-Jeong Kim (well@kfri.re.kr)
Hee-Joo Kang (anni0216@hanmail.net)
Hyang-Im Back (hyangim100@gmail.com)
Sun-Young Kim (sykim@jbctc.org)
Soo-Hyun Park (shpark@jbctc.org)
Hum-Young Baek (bhy0323@paran.com)
Yong-Jae Kim (yids06@nate.com)
Joon-Yeol Lee (jylee94@nate.com)
Soo-Wan Chae (soowan@jbnu.ac.kr)

Version: 2 Date: 30 July 2012

Author's response to reviews: see over
Cover letter

Dear Editors,

We wish to submit the attached manuscript entitled “Efficacy and Safety of Fermented Turmeric Powder in Subjects with Elevated Alanine Transaminase Levels” by Kim et al. for publication in "BMC-Complementary & Alternative Medicine". Our manuscript has been reviewed by several of our colleagues who have found it to be both scientifically sound and of general interest.

Thank you in advance for your consideration.

Sincerely yours,

Soo-Wan Chae, MD., PhD

Clinical Trial Center for Functional Foods,
Chonbuk National University Hospital,
Jeonju, Chonbuk 560-180,
Republic of Korea
TEL: 82-63-250-2348,  Fax: 82-63-250-2349
E-mail address: soowan@jbnu.ac.kr